Author's response to reviews

Title: A polymorphism at codon 31 of gene p21 is not associated with primary open angle glaucoma in Caucasians

Authors:

Thomas Ressiniotis (tomres@doctors.org.uk)
Philip G Griffiths (p.g.griffiths@ncl.ac.uk)
Sharon M Keers (s.m.keers@ncl.ac.uk)
Patrick F Chinnery (p.f.chinnery@ncl.ac.uk)
Michael Birch (birchmk@aol.com)

Version: 3 Date: 30 March 2005

Author's response to reviews: see over
Dear Editors,

MS: 9327328385915201

Thank you for accepting our manuscript for publication in your journal. The requested formatting changes have been made as below and the revised manuscript is attached.

Thank you for your attention.

Yours faithfully

Thomas Ressiniotis

Formatting changes

1. The title has been changed to: “A polymorphism at codon 31 of gene p21 is not associated with primary open angle glaucoma in Caucasians”.
2. “United kingdom” has been changed to “UK”.
3. The abstract has been structured as directed.
4. The manuscript sections have been structured as requested and all headings are now in sentence case.
6. References: throughout the entire manuscript a space has been created between the word and the reference.
7. Tables: Tables have been formatted in grid and moved after the references.